Author's response to reviews

Title: Age-related increases in parathyroid hormone may be antecedent to both osteoporosis and dementia

Authors:

Eric R Braverman (pathmedical@aol.com)
Thomas JH Chen (tjhchen@yahoo.com.tw)
Amanda LC Chen (tjhchen@yahoo.com.tw)
Thomas J Prihoda (PRIHODAT@uthscsa.edu)
Vanessa Arcuri (vanessaarcuri@yahoo.com)
Mallory M Kerner (mallory.kerner@gmail.com)
Anish Bajaj (dranishbajaj@gmail.com)
Javier Carbajal (javi64shoebox@aol.com)
Dasha Braverman (dasha@totalhealthnutrients.com)
B William Downs (bildowns001@comcast.net)
Kenneth Blum (drd2gene@aol.com)

Version: 8 Date: 2 August 2009

Author's response to reviews:

The author Thomas Prihoda has been removed. The University of Florida was added as an affiliation of Dr. Kenneth Blum. The statistical assistance of Dr. Pons was acknowledged in the acknowledgement section.